Author’s response to reviews

Title: Bridging gaps in everyday life - a free-listing approach to explore the variety of activities performed by physiotherapists in specialized palliative care

Authors:

Ulrika Olsson Möller (ulrika.olsson_moller@med.lu.se)
Kjerstin Stigmar (kjerstin.stigmar@med.lu.se)
Ingela Beck (ingela.beck@hkr.se)
Marlene Malmström (marlene.malmstrom@med.lu.se)
Birgit Rasmussen (birgit.rasmussen@med.lu.se)

Version: 2 Date: 12 Jan 2018

Author’s response to reviews:

Der mr Yau,

Thank you, we greatly appreciate your valuable comments. The following corrections are included in the uploaded version.

1. Please include the full name of the ethics committee (and the institute to which it belongs to) that approved the study and the committee’s reference number if appropriate.

If you did not need formal ethics approval please confirm that this complies with national guidelines and provide a reference which supports this. Alternatively, supply a statement that says that a local ethics committee ruled that no formal ethics approval was required in this particular case.

Our comment: The following text is included (p 18, line 424-425):

"...no formal ethical approval was requested which complies with national guidelines (51).


2. Please include in the title page the email addresses of all authors.

Our comment: The email addresses of all authors are included in the title page.
3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Our comment: A single, final, clean version of the manuscript and figure is uploaded.

Sincerely,

Ulrika Olsson Moller
Institute for Palliative Care, Lund, Sweden